

ESSAY

Rationale for WHO's New Position Calling for Prompt Reporting and Public Disclosure of Interventional Clinical Trial Results

Vasee S. Moorthy*, Ghassan Karam, Kirsten S. Vannice, Marie-Paule Kieny

World Health Organization, Geneva, Switzerland

* moorthyv@who.int

On April 14, 2015, the World Health Organization (WHO) published a new statement on the public disclosure of clinical trial results (<u>S1 Text</u>) [1]. The WHO statement not only re-affirms the ethical imperative of clinical trial results reporting, it also defines reporting timeframes, calls for results-reporting of older but still unpublished trials, and outlines steps to improve linkages between clinical trial registry entries and their published results. This updates and expands WHO's 2005 statement that "the registration of all interventional trials is a scientific, ethical, and moral responsibility" [2].

WHO's 2005 statement called for all interventional clinical trials to be registered. Subsequently, there has been an increase in clinical trial registration prior to the start of trials. This has enabled tracking of the completion and timeliness of clinical trial reporting. There is now a strong body of evidence showing failure to comply with results-reporting requirements across intervention classes, even in the case of large, randomised trials [3–7]. This applies to both industry and investigator-driven trials. In a study that analysed reporting from large clinical trials (over 500 participants) registered on clinicaltrials.gov and completed by 2009, 23% had no results reported even after a median of 60 months following trial completion; unpublished trials included nearly 300,000 participants [3]. Among randomised clinical trials (RCTs) of vaccines against five diseases registered in a variety of databases between 2006–2012, only 29% had been published in a peer-reviewed journal by 24 months following study completion [4]. At 48 months after completion, 18% of trials were not reported at all, which included over 24,000 participants. In another study, among 400 randomly selected clinical trials, nearly 30% did not publish the primary outcomes in a journal or post results to a clinical trial registry within four years of completion [5].

The Declaration of Helsinki and other statements have outlined the compelling reasons why interventional clinical trials should be reported in a timely fashion [8–10]. In brief, not reporting clinical trial results is likely to lead to dissemination bias. This bias has the following major adverse consequences:

- It affects understanding of the scientific state of the art.
- It leads to inefficiencies in resource allocation for both research and development and financing of health interventions.
- It creates indirect costs for public and private entities, including patients themselves, who pay for suboptimal or harmful treatments.
- It potentially distorts regulatory and public health decision making.



GOPEN ACCESS

Citation: Moorthy VS, Karam G, Vannice KS, Kieny M-P (2015) Rationale for WHO's New Position Calling for Prompt Reporting and Public Disclosure of Interventional Clinical Trial Results. PLoS Med 12(4): e1001819. doi:10.1371/journal.pmed.1001819

Published: April 14, 2015

Copyright: © 2015 Moorthy et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: There is no specific funding for this work.

Competing Interests: The authors have declared that no competing interests exist.

Provenance: Not commissioned; not externally peerreviewed Furthermore, it is unethical to conduct human research without publication and dissemination of the results of that research. In particular, withholding results may subject future volunteers to unnecessary risk.

The realisation of the importance of reporting results from clinical trials has led to the introduction of reporting requirements in an increasing number of jurisdictions. Requirements have been articulated in the current Declaration of Helsinki [8], by legislation in the United States [11], and at the European Union level [12]. The International Committee of Medical Journal Editors encourages posting of clinical trial results in clinical trial registries [13]. It has recently become mandatory for sponsors of clinical trials registered with the EU Clinical Trials Register to post results summaries on the registry database [12]. The US Department of Health and Human Services has proposed an expanded scope for clinical trials to be reported by law and clarified reporting requirements for posting results on clinicaltrials.gov. In addition, the US National Institutes of Health (NIH) is proposing to mandate clinical trials registration and results reporting for all NIH-funded clinical trials [14]. Initiatives such as Cochrane, AllTrials, and the OPEN Consortium (To Overcome failure to Publish nEgative fiNdings) are advocating strongly for greater transparency in results reporting. During 2014, one of the world's largest pharmaceutical companies adopted policies for public disclosure of clinical research that include posting of results from all clinical trials (Phase I-IV) on a free-to-access website and submission to a peer-reviewed journal in a short timeframe, going far beyond what was required by laws and regulations [15]. Despite the recent progress in some jurisdictions, gaps remain.

In its statement, WHO has outlined the need for reporting to occur in two modalities. The first is for the main findings of clinical trials to be submitted for publication in a peer-reviewed journal within 12 months of study completion (defined as the final data collection date for the primary outcome measure), with a further 12 months allowed from first submission to publication. Thus the key indicator for tracking will be journal publication of results within 24 months of study completion. Additionally, the key outcomes (defined in the statement) should be made publicly available within 12 months of study completion by posting to the results section of the primary clinical trial registry. If the registry does not allow posting, then the results should be posted on another easily accessible website.

Another important feature of the WHO statement is its call for public disclosure of results from older, unreported clinical trials. Results from past clinical trials still have an important bearing on scientific research today. Additional components of the WHO statement are a reaffirmation of registration of clinical trials before the first patient receives the medical intervention, and inclusion of a Trial ID (i.e., the clinical trial registry identifier) in the publication for easy linking of manuscripts with clinical trial registry entries. This will assist the ready identification of trials that have been conducted but not reported. It may be appropriate for funding agencies, particularly those spending taxpayer-sourced funds, to engage in such tracking as part of routine auditing for good use of research funds.

In 2005, WHO established the International Clinical Trials Registry Platform (ICTRP) to provide an overview of clinical trial research accessible to all those involved in health care decision making globally [2]. The ICTRP is a registry platform that regularly imports trial records from clinicaltrials.gov, ISRCTN, EU Clinical Trials Register, Australia New Zealand Clinical Trial Registry, Pan African Clinical Trial Registry, and Clinical Trial Registries from China, India, Brazil, Republic of Korea, Cuba, Germany, Iran, Japan, Sri Lanka, The Netherlands, and Thailand.

A current obstacle to reporting results is the lack of mechanisms on many clinical trial registries for sponsors to submit results for public access. Clinicaltrials.gov and the EU Clinical Trials Register are two registries that do so. Ensuring that all interventional clinical trials are reported will require action following the WHO statement. Data indicate that the extent of results-reporting has been insufficient to date and that incentives and legislation are needed to achieve compliance. WHO calls for ethics committees, regulatory authorities, professional bodies, sponsors, investigators, and funding agencies to act in their jurisdictions to ensure results from all interventional clinical trials are reported and publicly disclosed.

Supporting Information

S1 Text. WHO statement on public disclosure of clinical trial results. (PDF)

Author Contributions

Wrote the first draft of the manuscript: VSM KSV. Contributed to the writing of the manuscript: GK MPK. Agree with manuscript results and conclusions: VSM GK KSV MPK. ICMJE criteria for authorship read and met: VSM GK KSV MPK.

References

- 1. World Health Organization. WHO Statement on Public Disclosure of Clinical Trial Results. Accessed 14 April 2015. <u>http://www.who.int/ictrp/results/reporting</u>
- World Health Organization. International Clinical Trials Registry Platform (ICTRP). Accessed 18 February 2015. <u>www.who.int/ictrp</u>
- Jones CW, Handler L, Crowell KE, Keil LG, Weaver MA, Platts-Mills TF. Non-publication of large randomized clinical trials: cross sectional analysis. BMJ. 2013 Oct 29; 347:f6104. doi: <u>10.1136/bmj.f6104</u> PMID: <u>24169943</u>
- Manzoli L, Flacco ME, D'Addario M, Capasso L, De Vito C, Marzuillo C, et al. Non-publication and delayed publication of randomized trials on vaccines: survey. BMJ. 2014 May 16; 348:g3058. doi: <u>10.</u> <u>1136/bmj.g3058</u> PMID: <u>24838102</u>
- Saito H, Gill CJ. How frequently do the results from completed US clinical trials enter the public domain?—A statistical analysis of the ClinicalTrials.gov database. PLoS ONE. 2014 Jul 15; 9(7): e101826. doi: <u>10.1371/journal.pone.0101826</u> PMID: <u>25025477</u>
- Schmucker C, Schell LK, Portalupi S, Oeller P, Cabrera L, Bassler D, et al. Extent of non-publication in cohorts of studies approved by research ethics committees or included in trial registries. PLoS ONE. 2014 Dec 23; 9(12):e114023. doi: <u>10.1371/journal.pone.0114023</u> PMID: <u>25536072</u>
- Turner EH, Matthews AM, Linardatos E, Tell RA, Rosenthal R. Selective publication of antidepressant trials and its influence on apparent efficacy. N Engl J Med. 2008 Jan 17; 358(3):252–60. doi: <u>10.1056/</u> NEJMsa065779 PMID: 18199864
- World Medical Association. World Medical Association Declaration of Helsinki: Ethical Principles for Medical Research Involving Human Subjects. JAMA. 2013; 310(20):2191–2194. doi: <u>10.1001/jama.</u> <u>2013.281053</u> PMID: <u>24141714</u>
- Hudson KL, Collins FS. Sharing and reporting the results of clinical trials. JAMA. 2015 Jan 27; 313 (4):355–6. doi: <u>10.1001/jama.2014.10716</u> PMID: <u>25408371</u>
- Chalmers I. Underreporting research is scientific misconduct. JAMA. 1990 Mar 9; 263(10):1405–8. PMID: 2304220
- 11. Food and Drug Administration Amendments Act of 2007. Pub. L. No. 110–85, 121 Stat. 823 (September 27, 2007).
- 12. European Medicines Agency. Posting of clinical trial summary results in European Clinical Trials Database (EudraCT) to become mandatory for sponsors as of 21 July 2014. Accessed 18 February 2015. http://www.ema.europa.eu/ema/index.jsp?curl = pages/news_and_events/news/2014/06/news_ detail_002127.jsp&mid=WC0b01ac058004d5c1
- International Committee of Medical Journal Editors. Recommendations for the Conduct, Reporting, Editing, and Publication of Scholarly Work in Medical Journals—updated December 2014. Accessed 10 March 2015. <u>http://www.icmje.org/recommendations/</u>



- 14. US National Institutes of Health. HHS and NIH take steps to enhance transparency of clinical trial results. Accessed 18 February 2015. <u>http://www.nih.gov/news/health/nov2014/od-19.htm</u>
- GlaxoSmithKline. GSK Public policy positions: Public Disclosure of Clinical Research. Accessed 18 February 2015. <u>http://www.gsk.com/media/280827/disclosure-of-clinical-trial-information-policy.pdf</u>